Author's response to reviews

**Title:** Job satisfaction and retention of healthcare providers in Afghanistan and Malawi

**Authors:**

Linda Fogarty (linda.fogarty@jhpiego.org)
Young Mi Kim (young-mi.kim@jhpiego.org)
Hee-Soon Juon (hjuon@jhsph.edu)
Hannah Tappis (htappis@jhsph.edu)
Jin Won Noh (jinwon.noh@gmail.com)
Partamin Zainullah (partamin@jhpiego.org)
Aleisha Rozario (aleisha.rozario@jhpiego.org)

**Version:** 3  
**Date:** 23 December 2013

**Author's response to reviews:** see over
Human Resources for Health

To Whom It May Concern:

Please find attached a revised version of our manuscript, “Job satisfaction and retention of healthcare providers in Afghanistan and Malawi” for publication in Human Resources for Health.

As requested, the title has been changed from “Job satisfaction and retention of healthcare providers: a cross-sectional study in Afghanistan and Malawi”.

Best regards,

Linda Fogarty
Director of Monitoring, Evaluation and Research
Jhpiego – An affiliate of Johns Hopkins University
Baltimore, MD, USA
Email: linda.fogarty@jhpiego.org